ഹി



Geophysical Research Letters^{*}

RESEARCH LETTER

10.1029/2023GL103241

Key Points:

- Future emission pathways driven by climate actions are projected to alleviate surface ozone pollution in most parts of China by 2030
- However, for China's three main city clusters, model projections disagree strongly for two widely used chemical mechanisms
- This modeling uncertainty may arise from the inconsistency of categorizing ozone chemical regimes by different chemical mechanisms

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

X. Weng and J. Li, x.weng@uea.ac.uk; lijw@tea.ac.cn

Citation:

Weng, X., Li, J., Forster, G. L., & Nowack, P. (2023). Large modeling uncertainty in projecting decadal surface ozone changes over city clusters of China. *Geophysical Research Letters*, 50, e2023GL103241. https://doi. org/10.1029/2023GL103241

Received 10 FEB 2023 Accepted 15 APR 2023

Author Contributions:

Conceptualization: Xiang Weng, Jiawei Li, Peer Nowack Data curation: Xiang Weng Formal analysis: Xiang Weng Funding acquisition: Jiawei Li Investigation: Xiang Weng, Jiawei Li, Grant L. Forster, Peer Nowack Methodology: Xiang Weng, Jiawei Li Project Administration: Peer Nowack Resources: Grant L. Forster, Peer Nowack Software: Xiang Weng, Jiawei Li

© 2023 The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited and is not used for commercial purposes.

Large Modeling Uncertainty in Projecting Decadal Surface Ozone Changes Over City Clusters of China

Xiang Weng¹, Jiawei Li², Grant L. Forster^{1,3}, and Peer Nowack^{1,4,5}

¹School of Environmental Sciences, University of East Anglia, Norwich, UK, ²Key Laboratory of Regional Climate-Environment for Temperate East Asia, Institute of Atmospheric Physics, Chinese Academy of Sciences, Beijing, China, ³National Centre for Atmospheric Science, University of East Anglia, Norwich, UK, ⁴Climatic Research Unit, University of East Anglia, Norwich, UK, ⁵Institute of Theoretical Informatics, Karlsruhe Institute of Technology, Karlsruhe, Germany

Abstract Climate policies will affect future surface ozone pollution in China. Here, we simulate changes in summertime ozone across China by 2030 under four emission scenarios reflecting different levels of climate action. We also contrast results obtained with two different chemical mechanisms employed in the chemical transport model (WRF-Chem). With emission reductions in ozone precursors introduced by climate policies, both mechanisms show promising ozone mitigation for most parts of China. However, they disagree starkly in China's three main city clusters, where one mechanism projects worsening ozone pollution by 2030 despite the emission reductions. We analyze possible drivers of this important discrepancy, in particular the role of varying ozone chemical regimes affecting its sensitivity to emission changes. We recommend an intercomparison project to examine this critical modeling uncertainty among other models/mechanisms, which would be invaluable for informing local and regional emission control strategies that are based on single-model results.

Plain Language Summary Surface ozone pollution is harmful to both human health and ecosystems. Reducing ozone formation through effective emission control strategies has therefore been identified as a pressing need. Chemical transport models (CTMs) are important tools that can help scientists and policymakers assess how effectively the emission reductions may alleviate ozone pollution. However, we show that the predicted effectiveness of emission control strategies for ozone mitigation in areas within the three city clusters of China are strongly dependent on the choice of chemical mechanism commonly employed in CTMs. For example, given emission reductions driven by ambitious climate action, we find that projected ozone pollution in these regions could be improved or worsened by the year 2030 depending on the model mechanism used. Our work underlines the importance of considering and understanding this disagreement when it comes to projecting even near-term emission-control strategies. Furthermore, we highlight the potential benefits of conducting a multi-model/mechanism intercomparison project to better understand how and why different models/mechanisms disagree on the simulated ozone response to emission changes, as to produce more robust mitigation scenario assessments.

1. Introduction

Surface ozone pollution is one of the key environmental concerns in China. In contrast to the remarkable reduction in fine particle ($PM_{2.5}$) pollution driven by clean air policies (Zhang et al., 2019), many studies report a worsening of ozone pollution in urban regions of China over the last decade (Li et al., 2020; Y. Liu & Wang, 2020a; X. Lu et al., 2018, 2020; Weng et al., 2022). Earlier worsening ozone pollution, from 2013 to 2017, have been attributed to ozone's nonlinear response to large reductions in nitrogen oxides (NO_x) emissions without equivalent reductions in volatile organic compounds (VOCs) emissions (Y. Liu & Wang, 2020b; N. Wang et al., 2019; T. Wang et al., 2017), particularly in many urban regions that were within the NO_x -saturated regime (H. Lu et al., 2019; Ou et al., 2016). Nevertheless, with the continuous reductions in NO_x , the sensitivity of summertime ozone production to VOCs in some urban areas of China may have gradually weakened (W. Wang et al., 2021, 2022). Furthermore, it has been suggested that the turning point between NO_x -saturated and NO_x -limited regimes in some densely populated urban areas of China was reached in 2019 (Chen et al., 2021).

A key question is how ozone pollution in China can be effectively mitigated through future emission controls, especially considering potential co-benefits from policy measures aimed at climate change mitigation. For



L. Forster, Peer Nowack Visualization: Xiang Weng

Nowack

Nowack

Supervision: Grant L. Forster, Peer

Writing - original draft: Xiang Weng

Weng, Jiawei Li, Grant L. Forster, Peer

Writing - review & editing: Xiang

instance, in 2016, China committed its Nationally Determined Contribution (NDC) pledges within the context of Paris Agreement; and further scaled up its commitment in 2020, by proposing the goal of achieving carbon Validation: Xiang Weng, Jiawei Li, Grant neutrality (Cheng et al., 2021; Z. Liu et al., 2022). This ambitious climate action may bring large emission reductions in CO_2 accompanied by reductions in emissions of other pollutants such as $PM_{2,5}$ and ozone precursors, which may lead to air quality improvements in the near future. For example, Cheng et al. (2021) suggested that by 2030, following the emission reduction plans in NDC pledges, the majority of the Chinese population may be exposed to less PM_{2.5} pollution ($<35 \ \mu g \cdot m^{-3}$). For surface ozone, long-term projections under the Representative Concentration Pathways (RCPs) using chemical transport models (CTMs) have been conducted by previous studies (Hong et al., 2019; Y. Wang, Hu, et al., 2021; Zhu & Liao, 2016) but fewer studies (e.g., Shi et al., 2021) have examined the effectiveness of emission controls on ozone mitigation following recent carbon neutrality plan which reflects China's up-to-date emission control strategies. Moreover, projections by these studies were often heavily dependent on the results from a single CTM. This presents a limiting factor as varying uncertainties and configurations in model setup can produce inconsistent predictions (Gilliam et al., 2015; Thomas et al., 2019). For instance, it has been well documented that the choice of chemical mechanism in a model can lead to discrepancies in simulated gaseous species and aerosols (Archibald et al., 2020; Balzarini et al., 2015; Crippa et al., 2019; Mar et al., 2016; Visser et al., 2019; Yang et al., 2018). Consequently, disagreement in ozone projections from different chemical mechanisms can ultimately affect policymaking for ozone mitigation. Nevertheless, this issue has not been considered in sufficient detail in recent CTM studies that assess the effectiveness of current or future nationwide ozone mitigation strategies in China.

> Therefore, we here conduct simulations with two widely used chemical mechanisms employed in a state-of-art CTM. We highlight and discuss noticeable discrepancies in ozone projections when considering near-future emission reductions driven by climate actions.

2. Materials and Methods

2.1. Model Setup

We use the Weather Research and Forecasting Model with Chemistry (WRF-Chem; Grell et al., 2005) standard version 4.1.5 to conduct simulations for the boreal summer period (i.e., June, July, and August). The simulation domain covers the entirety of China (see Figure S1 in Supporting Information S1). The model setup is based on the settings of Silver et al. (2020). The details of the setup are documented in Text S1 of Supporting Information **S1**.

To simulate gas phase chemistry, we use WRF-Chem in two configurations (i.e., chemical mechanisms) with (a) the Model for Ozone and Related Chemical Tracers (MOZART; Emmons et al., 2010) and (b) the Carbon Bond Mechanism Z (CBMZ; Zaveri & Peters, 1999). Aerosols are simulated using a 4-bin Model for Simulating Aerosol Interactions and Chemistry (MOSAIC; Zaveri et al., 2008) which is coupled to these two gas-phase mechanisms (i.e., MOZART and CBMZ) in WRF-Chem. Biogenic emissions for both mechanisms are calculated online by the Model of Emissions of Gases and Aerosol from Nature (MEGAN; Guenther et al., 2000). Biomass burning emission is not included in all the simulation runs, following Crippa et al. (2019), in order to avoid the inconsistence in processing fire emission between MOZART and CBMZ. Simulations by both chemical mechanisms share the same settings of physics and dynamics.

2.2. Simulations of Emission Scenarios

A total of five scenarios are simulated. These include a base simulation for which emissions are set at the level of summer 2017 to represent a real-world scenario. The remaining four simulations project ozone under different emission pathways for the summer of 2030. All scenarios are run for both chemical mechanisms.

The emission data used in the base simulation are from Multi-resolution Emission Inventory for China (MEIC) version 1.3, and we denote the base simulation as "Base-2017" hereafter. For model evaluation, we compare the simulated pollutants from Base-2017 with observational data (see Text S2 in Supporting Information S1). For simulations with emissions projected for summer 2030, data from the Dynamic Projection model for Emissions in China (DPEC) version 1.1 (Cheng et al., 2021) are used. The development of DPEC is based on MEIC (Tong et al., 2020).

Table 1

Summary of the Emissions Scenarios and Their Corresponding Labels

	Change of NO, emissions relative to	Change of NMVOC emissions relative	
Emission scenario	2017	to 2017	Scenario label
MEIC emissions in 2017	_	_	Base-2017
Baseline emissions in 2030	41.0%	12.4%	Limited-controls-2030
Current-goals emissions in 2030	-41.5%	-23.2%	Current-goals-2030
Ambitious-pollution-Neutral-goals emissions in 2030	-60.2%	-28.9%	Neutral-goals-2030
Ambitious-pollution-1.5°C-goals emissions in 2030	-60.4%	-30.4%	1.5°C-goals-2030

For the future projections, we consider four emission pathways from DPEC. These include three projected emission reduction pathways from a moderate reduction scenario considering current released and upcoming emission control policies (i.e., "Current-goals") to the other two ambitious scenarios for the pursuit of carbon neutrality ("Ambitious-pollution-Neutral-goals") and 1.5° C temperature limit ("Ambitious-pollution-1.5°C-go als"). Besides, we also include an additional pathway named "Baseline" by Cheng et al. (2021), which has overall increases of emissions compared to 2017 levels as it represents limited actions in emission controls. To avoid confusion with "Base-2017" (i.e., the real-world scenario), we term simulations of "Baseline" as "Limited-controls-2030" herein. A more detailed description of all these emission pathways is presented in Text S3 in Supporting Information S1 and can be referred to Cheng et al. (2021). In Table 1, we list all scenarios and their labels together with their relative-to-2017 changes of summertime NO_x and non-methane VOC (NMVOC) total emissions in China. Overall, relative to 2017, the sign of emission changes (i.e., increases or decreases) are spatially consistent over eastern China in each 2030 scenario (Figure S8 in Supporting Information S1).

We compare results for each of the projected emission scenarios in 2030 with the Base-2017 levels. In particular, we are interested in how future scenarios may induce changes of summertime maximum daily 8 hr average (MDA8) ozone pollution relative to 2017 levels across China. To account for emissions from outside of China, EDGAR version 5.0 at 2015 levels (Mogno & Marvin, 2022) is used. These emissions from outside China, as well as other settings of WRF-Chem, initial and boundary conditions for both meteorology and chemistry (set at the summer of 2017, see Text S1 in Supporting Information S1 for details) and biogenic emissions, remain fixed for all of the simulations. This setup reflects our objective to isolate the effect of emission changes within China on its own ozone levels by 2030.

3. Results and Discussion

3.1. Projected Ozone Changes

Figure 1 shows projections of summertime ozone changes for 2030, relative-to-2017, for the two different chemical mechanisms.

Both mechanisms show overall agreement in predicting the sign of ozone changes in areas outside of the three highly populated city clusters which are Beijing–Tianjin–Hebei (BTH), Yangtze River Delta (YRD) and Pearl River Delta (PRD). Specifically, due to the lax control policies in Limited-controls-2030, ozone increases occur in these areas; whereas decreases in ozone are found under the three emission reduction pathways (Current-goals-2030, Neutral-goals-2030 and 1.5°C-goals-2030). As the ozone sensitivity in these areas have been mostly classified as NO_x-limited and transitional regime (e.g., Ren et al., 2022; W. Wang et al., 2021), it is not unexpected that ozone changes over these areas follow a similar trend as the changes in NO_x emissions (see Table 1 and Figure S8 in Supporting Information S1): lower NO_x emissions lead to ozone decreases and vice versa.

In contrast, in regions within the three city clusters where ozone sensitivity has been classified to be mostly in the NO_x -saturated regime (Li et al., 2019; Y. Liu & Wang, 2020b; Ren et al., 2022), even whether ozone may increase or decrease is disagreed by the two mechanisms. For example, in Current-goals-2030, despite the emissions of NO_x and NMVOC in these three city clusters being reduced when compared with the Base-2017 emissions (Figure S8 in Supporting Information S1), the surface ozone levels in some of these areas are predicted to be enhanced by 4–10 ppbv by MOZART (Figure 1b), lifting pollution levels further above the threshold of air quality guideline for short-term ozone exposure (100 μ g m⁻³ for MDA8 ozone, roughly 50 ppbv) proposed by World



Geophysical Research Letters

19448007, 2023, 9, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GL103241 by Test, Wiley Online Library on [02/05/2023]. See the Terms and Conditions

s (nups



Figure 1. Simulated summertime average maximum daily 8 hr average ozone changes for each emission scenario for the year 2030, relative to Base-2017 levels. Results are shown for the two WRF-Chem configurations, with the chemical mechanisms Model for Ozone and Related Chemical Tracers (a–d) and Carbon Bond Mechanism Z (e–h), respectively. The boundaries for the three city clusters, where ozone changes mostly do not agree on the sign, are highlighted and labeled in (a).

Health Organization. Moreover, most of these simulated increases persist despite stronger emission reductions in Neutral-goals-2030 and 1.5°C-goals-2030 (Figures 1c and 1d). In stark contrast, CBMZ predicts that ozone concentrations in these regions decrease or remain close to the 2017 levels under the three emission reduction pathways (Figures 1f–1h). Crucially, this would imply different requirements for policymakers to further control emissions of ozone precursors on top of emission reductions aimed at achieving already ambitious climate policy goals. To quantitatively summarize these results, we compare averaged ozone changes for the polluted urban and industrial regions within the three city clusters and the rest of China in Figure 2.





4 of 11

conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

Overall, this discrepancy highlights major modeling uncertainty in projecting future ozone changes over urban and industrial regions of China, with important implications for policymakers. Using CBMZ, one might conclude that emission controls following the three emission reduction pathways are promising for mitigating ozone pollution in most regions, including those within the city clusters. This agrees well with previous findings of Chen et al. (2021) and Kang et al. (2021). Using MOZART, however, would indicate that these emission reductions might remain insufficient in some of the most highly populated areas of China, where increasing ozone pollution would negatively affect a large portion of the Chinese population.

3.2. Discrepancy in Ozone Sensitivity

The striking discrepancies in the projected ozone changes may be mainly the result of differences in gas-phase chemistry for ozone production in MOZART and CBMZ, rather than being driven by aerosol effects (Text S4 in Supporting Information S1). A most likely cause is their different chemical sensitivities to NO_x and VOC emission changes. To illustrate this, we conducted a set of test simulations for July 2017 in which the emissions across China of NO_x and VOCs were reduced by 10%, 30%, and 50%, respectively.

In essence, these simulations underline the important differences in how ozone responds within the three city clusters to a given change in emissions, in particular for reduced NO_x emissions (Figure 3). While both mechanisms show decreases in ozone when VOC emissions are reduced, consistent with the simulations by Kang et al. (2021), ozone in NO_x -saturated urban and industrial regions exhibits strikingly different responses to the perturbation of NO_x emissions. For instance, MOZART shows that the MDA8 ozone increases on average by over 5 ppbv in the urban and industrial regions when NO_x emissions are reduced by 30% or even 50% (Figure 3a). However, for CBMZ, decreases in ozone are predicted given the same NO_x emission reductions. In Addition, the ozone increases following 30% NO_x emission reductions are more pronounced and more spatially extended in MOZART (Figure 3c) when compared to CBMZ (Figure 3e), analogous to the results for emission reductions in 2030 (Figure 1). This suggests that the difference in how ozone responds to NO_x reductions is the key factor to explain the mechanism dependency.

Our tests here also indicate a disagreement in categorizing ozone chemical regimes in urban and industrial regions: the results are consistent with MOZART tending toward a more NO_x -saturated regime whereas a transitional or NO_x -limited regime may be suggested by CBMZ. Similarly, an earlier study by Knote et al. (2015) showed that simulations using MOZART tends to predict more air quality stations within NO_x -saturated regime in the North America than using CBMZ. Our results indicate that this phenomenon may also occur in China, where alleviating ozone pollution is currently pivotal for promoting public health. Such important modeling uncertainties, and their implications, should ideally be considered during the decision-making and design of future ozone control strategies.

3.3. Possible Causes of the Chemical Regime Discrepancy

Increasing ozone pollution given reductions in NO_x emissions is not unexpected in a NO_x-saturated regime. First, ozone can be consumed through its reaction with NO, which is known as NO_x-titration (i.e., NO + O₃ \rightarrow NO₂ + O₂) (Sillman, 1999). With reductions in NO_x emissions, NO_x-titration can be weakened, thereby leading to less ozone suppression. Second, reducing NO_x emissions in an NO_x-polluted environment can weaken the formation of nitric acid (HNO₃) that can be formed through a chain terminating reaction with the hydroxyl radical (OH) (OH + NO₂ \rightarrow HNO₃). In this scenario, this reaction becomes a major sink of HO_x (HO_x \equiv OH + H + preoxy radicals; Jacob, 2000), which affects VOCs oxidation, thus suppressing subsequent net ozone production. Reducing NO_x emissions in such a scenario leads to more available OH for atmospheric oxidation and can ultimately enhance ozone formation (H. Lu et al., 2019; Sillman, 1999).

Mar et al. (2016) showed that inconsistencies in rate constants of inorganic gas phase reactions among different mechanisms can cause a large discrepancy in simulated ozone concentrations, which may also result in different ozone sensitivity regimes. We indeed find that the MOZART mechanism has higher rate constants for both NO_x -titration and HNO_3 formation under usual atmospheric conditions (see Text S5 in Supporting Information S1). This suggests that MOZART may have a faster suppression of net ozone formation. In turn, ozone increases may also be more responsive to NO_x reductions than for CBMZ. We tested this hypothesis by setting the rate constants for NO_x -titration and HNO_3 formation (or rate constants for most of the other inorganic reactions that are listed by Mar et al. (2016)) the same for both mechanisms. However, the discrepancy in ozone changes





Figure 3. Averaged maximum daily 8 hr average ozone changes in response to the emission reductions of NO_x and volatile organic compounds (VOCs) during July 2017 in urban and industrial regions within the three city clusters (a) and the rest of China (b). Spatial patterns of ozone changes in Model for Ozone and Related Chemical Tracers given NO_x (c) and VOCs (d) emissions reductions of 30%. The same, but for Carbon Bond Mechanism Z (e, f).





Figure 4. Differences of daytime (06:00–18:00) average concentrations of NO_x (a), HNO₃ (b), OH (c), and O₃ (d) during July in the Base-2017 scenario between Carbon Bond Mechanism Z (CBMZ) and Model for Ozone and Related Chemical Tracers (MOZART) (i.e., predicted concentration levels in CBMZ subtracted by MOZART).

persisted, suggesting that the differences in these rate constants alone cannot explain the predicted discrepancy. Clearly, rates of chemical reactions are not only determined by the rate constant but will also depend on the reactant concentrations. The simulated concentrations of key chemical compounds involved in NO_x-titration and HNO₃ formation indeed also differ significantly for the two mechanisms. For example, in Base-2017, the simulated NO_x are lower in CBMZ (Figure 4a) despite the fact that both of the mechanisms have the same emissions of NO_x, suggesting that there is a stronger sink for NO_x (e.g., HNO₃ formation) in CBMZ. This is also consistent with the higher concentrations of simulated HNO₃ when using the CBMZ mechanism (Figure 4b). It is likely that the higher simulated abundance of OH in CBMZ (Figure 4c) may be an important contributing factor that leads to a stronger formation of HNO₃, even when considering its lower reaction rate constant than the one in MOZART. Higher OH can therefore lead to higher ozone concentrations (Figure 4d) and vice versa.

In summary, with respect to NO_x -titration, MOZART may have an overall stronger NO_x -titration as it has higher simulated NO_x (i.e., weaker sink for NO_x ; see Figure 4a) and a higher rate constant for NO_x -titration (Text S5 in

Supporting Information S1). Given this, the simulated ozone increases in MOZART may be more sensitive to a reduction in NO_x emission. Furthermore, the higher levels of simulated OH (Figure 4c) in CBMZ may already be sufficient to drive more ozone formation (Figure 4d), and higher ozone also leads to more OH. If this is the case, then the addition of OH from the reduced HNO₃ formation induced by NO_x reduction might not have a large effect on ozone increases in CBMZ. Therefore, with the combined effect of all these factors, simulated ozone increases may be less pronounced in CBMZ. We further hypothesize that ozone changes in CBMZ might move closer to the simulated increases by MOZART, if additional heterogenous reactions or aerosol uptakes of HO_x and ozone (e.g., Y. Liu & Wang, 2020a) are included.

Overall, our results suggest that a more thorough intercomparison of NO_x -titration, HNO_3 formation along with their corresponding chemical species across several chemical mechanisms is necessary for elucidating the discrepancy in simulated ozone responses to near-term or future emission changes. We further note that soil NO_x emissions could also be addressed in future intercomparisons, because such emissions may affect ozone responses to anthropogenic emissions changes (X. Lu et al., 2021; Y. Wang, Fu, et al., 2021).

4. Conclusions

Climate-action-driven emission controls will substantially change surface ozone pollution in China. Reliable projections of ozone changes given emission reductions in ozone precursors are key to assess which control strategy may be optimal for future ozone mitigation. Here, we show that two widely used chemical mechanisms from a CTM could produce highly inconsistent conclusions on the efficacy of planned control measures. Specifically, we project ozone changes under four climate-action-driven emission pathways by summer 2030 using two chemical mechanisms (i.e., MOZART and CBMZ) in WRF-Chem. Although both mechanisms agree that summertime ozone in most parts of China can be mitigated to lower-than-2017 levels given emission reductions, we find marked discrepancies in major populated city clusters of China. In particular, MOZART simulates worsening ozone pollution in some of these areas by 2030 despite the ambitious emission reductions in ozone precursors as part of the proposed climate actions. In contrast, CBMZ typically shows reduced ozone pollution for the same scenarios and areas. We propose that this opposite response can mainly be attributed to differences in the simulated ozone chemical regimes. Policy-making aimed at ozone mitigation that often relies on model simulations should be aware of such major discrepancies, because it can lead to inconsistent conclusions regarding the effectiveness of emission control strategies. Therefore, we see an urgent need for a multi-model intercomparison project involving various chemical mechanisms from CTMs in order to achieve a more thorough understanding of how and why simulated ozone in different mechanisms/models responds differently to emission changes of its precursors.

Data Availability Statement

The source code of WRF-Chem version 4.1.5 is available at https://github.com/wrf-model/WRF/releases/tag/ v4.1.5. MEIC (version 1.3) emission data can be accessed from http://meicmodel.org.cn/?page_id=541&lang=en, and DPEC emission data (version 1.1) are available at http://meicmodel.org.cn/?page_id=1918&lang=en. EDGAR emission version 5.0 can be downloaded from https://zenodo.org/record/6130621. Meteorological data from National Centers for Environmental Prediction (NCEP) Global Data Assimilation System (GDAS)/final analysis (FNL) are provided at https://rda.ucar.edu/datasets/ds083.3/. Simulations by Community Atmosphere Model with Chemistry (CAM-chem) can be accessed from https://www.acom.ucar.edu/cam-chem/cam-chem. shtml. Surface hourly measurement data of O_3 , NO_2 and $PM_{2.5}$ provided by the Chinese Ministry of Ecology and Environment (MEE) are available at https://zenodo.org/record/7629985. Model results and configuration files for WRF-Chem simulations can be downloaded from https://zenodo.org/record/7625666. WRF-Chem preprocessing tools mozbc, anthro_emiss and bio_emiss can be downloaded from https://www.acom.ucar.edu/wrf-chem/ download.shtml.

References

Archibald, A. T., Turnock, S. T., Griffiths, P. T., Cox, T., Derwent, R. G., Knote, C., & Shin, M. (2020). On the changes in surface ozone over the twenty-first century: Sensitivity to changes in surface temperature and chemical mechanisms: 21st century changes in surface ozone.

Jiawei Li was supported by National Key R&D Program of China (2022YFF0802503). We thank MEIC team for granting us access to both MEIC and DPEC data. Model simulations were carried out on the High Performance Computing Cluster supported by the Research and Specialist Computing Support service at the University of East Anglia. We acknowledge use of WRF-Chem preprocessing tools mozbc, anthro emiss and bio emiss. We thank Wencong Chen from Chinese Academy of Meteorological Sciences and Lei Chen from Nanjing University of Information Science & Technology for their help in validating our WRF-Chem simulations

and-conditions

on Wiley Online Library for

of use; OA articles are governed by the applicable Creative Commons License

and Conditions

19448007, 2023, 9, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GL103241 by Test, Wiley Online Library on [02/05/2023]. See the Terms

Philosophical Transactions of the Royal Society A: Mathematical, Physical and Engineering Sciences, 378(2183), 20190329. https://doi.org/10.1098/rsta.2019.0329

- Balzarini, A., Pirovano, G., Honzak, L., Žabkar, R., Curci, G., Forkel, R., et al. (2015). WRF-Chem model sensitivity to chemical mechanisms choice in reconstructing aerosol optical properties. Atmospheric Environment, 115, 604–619. https://doi.org/10.1016/j.atmosenv.2014.12.033
- Chen, X., Jiang, Z., Shen, Y., Li, R., Fu, Y., Liu, J., et al. (2021). Chinese regulations are working—Why is surface ozone over industrialized areas still high? Applying lessons from Northeast US air quality evolution. *Geophysical Research Letters*, 48(14), 1–9. https://doi. org/10.1029/2021GL092816
- Cheng, J., Tong, D., Zhang, Q., Liu, Y., Lei, Y., Yan, G., et al. (2021). Pathways of China's PM_{2.5} air quality 2015–2060 in the context of carbon neutrality. *National Science Review*, 8(12), nwab078. https://doi.org/10.1093/nsr/nwab078
- Crippa, P., Sullivan, R. C., Thota, A., & Pryor, S. C. (2019). Sensitivity of simulated aerosol properties over eastern North America to WRF-Chem parameterizations. *Journal of Geophysical Research: Atmospheres*, 124(6), 3365–3383. https://doi.org/10.1029/2018JD029900
- Emmons, L. K., Walters, S., Hess, P. G., Lamarque, J. F., Pfister, G. G., Fillmore, D., et al. (2010). Description and evaluation of the Model for Ozone and Related chemical Tracers, version 4 (MOZART-4). *Geoscientific Model Development*, 3(1), 43–67. https://doi.org/10.5194/ gmd-3-43-2010
- Gilliam, R. C., Hogrefe, C., Godowitch, J. M., Napelenok, S., Mathur, R., & Rao, S. T. (2015). Impact of inherent meteorology uncertainty on air quality model predictions. *Journal of Geophysical Research: Atmospheres*, 120(23), 12259–12280. https://doi.org/10.1002/2015JD023674
- Grell, G. A., Peckham, S. E., Schmitz, R., McKeen, S. A., Frost, G., Skamarock, W. C., & Eder, B. (2005). Fully coupled "online" chemistry within the WRF model. Atmospheric Environment, 39(37), 6957–6975. https://doi.org/10.1016/j.atmosenv.2005.04.027
- Guenther, A., Geron, C., Pierce, T., Lamb, B., Harley, P., & Fall, R. (2000). Natural emissions of non-methane volatile organic compounds, carbon monoxide, and oxides of nitrogen from North America. *Atmospheric Environment*, 34(12–14), 2205–2230. https://doi.org/10.1016/ \$1352-2310(99)00465-3
- Hong, C., Zhang, Q., Zhang, Y., Davis, S. J., Tong, D., Zheng, Y., et al. (2019). Impacts of climate change on future air quality and human health in China. Proceedings of the National Academy of Sciences of the United States of America, 116(35), 17193–17200. https://doi.org/10.1073/ pnas.1812881116
- Jacob, D. J. (2000). Heterogeneous chemistry and tropospheric ozone. Atmospheric Environment, 34(12–14), 2131–2159. https://doi.org/10.1016/ S1352-2310(99)00462-8
- Kang, M., Zhang, J., Zhang, H., & Ying, Q. (2021). On the relevancy of observed ozone increase during COVID-19 lockdown to summertime ozone and PM_{2.5} control policies in China. *Environmental Science and Technology Letters*, 8(4), 289–294. https://doi.org/10.1021/acs. estlett.1c00036
- Knote, C., Tuccella, P., Curci, G., Emmons, L., Orlando, J. J., Madronich, S., et al. (2015). Influence of the choice of gas-phase mechanism on predictions of key gaseous pollutants during the AQMEII phase-2 intercomparison. Atmospheric Environment, 115, 553–568. https://doi. org/10.1016/j.atmosenv.2014.11.066
- Li, K., Jacob, D. J., Liao, H., Shen, L., Zhang, Q., & Bates, K. H. (2019). Anthropogenic drivers of 2013–2017 trends in summer surface ozone in China. Proceedings of the National Academy of Sciences of the United States of America, 116(2), 422–427. https://doi.org/10.1073/ pnas.1812168116
- Li, K., Jacob, D. J., Shen, L., Lu, X., De Smedt, I., & Liao, H. (2020). Increases in surface ozone pollution in China from 2013 to 2019: Anthropogenic and meteorological influences. Atmospheric Chemistry and Physics, 20(19), 11423–11433. https://doi.org/10.5194/acp-20-11423-2020
- Liu, Y., & Wang, T. (2020a). Worsening urban ozone pollution in China from 2013 to 2017 Part 1: The complex and varying roles of meteorology. Atmospheric Chemistry and Physics, 20(11), 6305–6321. https://doi.org/10.5194/acp-20-6305-2020
- Liu, Y., & Wang, T. (2020b). Worsening urban ozone pollution in China from 2013 to 2017—Part 2: The effects of emission changes and implications for multi-pollutant control. Atmospheric Chemistry and Physics, 20(11), 6323–6337. https://doi.org/10.5194/acp-20-6323-2020
- Liu, Z., Deng, Z., He, G., Wang, H., Zhang, X., Lin, J., et al. (2022). Challenges and opportunities for carbon neutrality in China. Nature Reviews Earth and Environment, 3(2), 141–155. https://doi.org/10.1038/s43017-021-00244-x
- Lu, H., Lyu, X., Cheng, H., Ling, Z., & Guo, H. (2019). Overview on the spatial-temporal characteristics of the ozone formation regime in China. Environmental Science: Processes and Impacts, 21(6), 916–929. https://doi.org/10.1039/c9em00098d
- Lu, X., Hong, J., Zhang, L., Cooper, O. R., Schultz, M. G., Xu, X., et al. (2018). Severe surface ozone pollution in China: A global perspective. Environmental Science and Technology Letters, 5(8), 487–494. https://doi.org/10.1021/acs.estlett.8b00366
- Lu, X., Ye, X., Zhou, M., Zhao, Y., Weng, H., Kong, H., et al. (2021). The underappreciated role of agricultural soil nitrogen oxide emissions in ozone pollution regulation in North China. *Nature Communications*, 12(1), 5021. https://doi.org/10.1038/s41467-021-25147-9
- Lu, X., Zhang, L., Wang, X., Gao, M., Li, K., Zhang, Y., et al. (2020). Rapid increases in warm-season surface ozone and resulting health impact in China since 2013. Environmental Science and Technology Letters, 7(4), 240–247. https://doi.org/10.1021/acs.estlett.0c00171
- Mar, K. A., Ojha, N., Pozzer, A., & Butler, T. M. (2016). Ozone air quality simulations with WRF-Chem (v3.5.1) over Europe: Model evaluation and chemical mechanism comparison. *Geoscientific Model Development*, 9(10), 3699–3728. https://doi.org/10.5194/gmd-9-3699-2016
- Mogno, C., & Marvin, M. R. (2022). EDGAR v5.0 emissions inventory speciated for the MOZART chemical mechanism. Zenodo. https://doi.org/10.5281/zenodo.6130621
- Ou, J., Yuan, Z., Zheng, J., Huang, Z., Shao, M., Li, Z., et al. (2016). Ambient ozone control in a photochemically active region: Short-term despiking or long-term attainment? *Environmental Science and Technology*, 50(11), 5720–5728. https://doi.org/10.1021/acs.est.6b00345
- Ren, J., Guo, F., & Xie, S. (2022). Diagnosing ozone–NOx–VOC sensitivity and revealing causes of ozone increases in China based on 2013– 2021 satellite retrievals. Atmospheric Chemistry and Physics, 22(22), 15035–15047. https://doi.org/10.5194/acp-22-15035-2022
- Shi, X., Zheng, Y., Lei, Y., Xue, W., Yan, G., Liu, X., et al. (2021). Air quality benefits of achieving carbon neutrality in China. Science of the Total Environment, 795, 148784. https://doi.org/10.1016/j.scitotenv.2021.148784
- Sillman, S. (1999). The relation between ozone, NOx and hydrocarbons in urban and polluted rural environments. *Atmospheric Environment*, 33(12), 1821–1845. https://doi.org/10.1016/S1352-2310(98)00345-8
- Silver, B., Conibear, L., Reddington, C. L., Knote, C., Arnold, S. R., & Spracklen, D. V. (2020). Pollutant emission reductions deliver decreased PM_{2.5}-caused mortality across China during 2015–2017. Atmospheric Chemistry and Physics, 20(20), 11683–11695. https://doi.org/10.5194/acp-20-11683-2020
- Thomas, A., Huff, A. K., Hu, X. M., & Zhang, F. (2019). Quantifying uncertainties of ground-level ozone within WRF-Chem simulations in the mid-Atlantic region of the United States as a response to variability. *Journal of Advances in Modeling Earth Systems*, 11(4), 1100–1116. https://doi.org/10.1029/2018MS001457
- Tong, D., Cheng, J., Liu, Y., Yu, S., Yan, L., Hong, C., et al. (2020). Dynamic projection of anthropogenic emissions in China: Methodology and 2015-2050 emission pathways under a range of socio-economic, climate policy, and pollution control scenarios. Atmospheric Chemistry and Physics, 20(9), 5729–5757. https://doi.org/10.5194/acp-20-5729-2020

- Visser, A. J., Folkert Boersma, K., Ganzeveld, L. N., & Krol, M. C. (2019). European NOx emissions in WRF-Chem derived from OMI: Impacts on summertime surface ozone. Atmospheric Chemistry and Physics, 19(18), 11821–11841. https://doi.org/10.5194/acp-19-11821-2019
- Wang, N., Lyu, X., Deng, X., Huang, X., Jiang, F., & Ding, A. (2019). Aggravating O3 pollution due to NOx emission control in eastern China. Science of the Total Environment, 677(x), 732–744. https://doi.org/10.1016/j.scitotenv.2019.04.388
- Wang, T., Xue, L., Brimblecombe, P., Lam, Y. F., Li, L., & Zhang, L. (2017). Ozone pollution in China: A review of concentrations, meteorological influences, chemical precursors, and effects. Science of the Total Environment, 575, 1582–1596. https://doi.org/10.1016/j. scitotenv.2016.10.081
- Wang, W., Parrish, D. D., Wang, S., Bao, F., Ni, R., Li, X., et al. (2022). Long-term trend of ozone pollution in China during 2014–2020: Distinct seasonal and spatial characteristics and ozone sensitivity. Atmospheric Chemistry and Physics, 22(13), 8935–8949. https://doi.org/10.5194/ acp-22-8935-2022
- Wang, W., van der A, R., Ding, J., Van Weele, M., & Cheng, T. (2021). Spatial and temporal changes of the ozone sensitivity in China based on satellite and ground-based observations. Atmospheric Chemistry and Physics, 21(9), 7253–7269. https://doi.org/10.5194/acp-21-7253-2021
- Wang, Y., Fu, X., Wu, D., Wang, M., Lu, K., Mu, Y., et al. (2021). Agricultural fertilization aggravates air pollution by stimulating soil nitrous acid emissions at high soil moisture. *Environmental Science and Technology*, 55(21), 14556–14566. https://doi.org/10.1021/acs.est.1c04134
- Wang, Y., Hu, J., Zhu, J., Li, J., Qin, M., Liao, H., et al. (2021). Health Burden and economic impacts attributed to PM_{2.5} and O₃ in China from 2010 to 2050 under different representative concentration pathway scenarios. *Resources, Conservation and Recycling*, 173, 105731. https:// doi.org/10.1016/j.resconrec.2021.105731
- Weng, X., Forster, G. L., & Nowack, P. (2022). A machine learning approach to quantify meteorological drivers of ozone pollution in China from 2015 to 2019. Atmospheric Chemistry and Physics, 22(12), 8385–8402. https://doi.org/10.5194/acp-22-8385-2022
- Yang, J., Kang, S., & Ji, Z. (2018). Sensitivity analysis of chemical mechanisms in the WRF-Chem model in reconstructing aerosol concentrations and optical properties in the Tibetan Plateau. Aerosol and Air Quality Research, 18(2), 505–521. https://doi.org/10.4209/aaqr.2017.05.0156
- Zaveri, R. A., Easter, R. C., Fast, J. D., & Peters, L. K. (2008). Model for simulating aerosol Interactions and chemistry (MOSAIC). Journal of Geophysical Research, 113(13), 1–29. https://doi.org/10.1029/2007JD008782
- Zaveri, R. A., & Peters, L. K. (1999). A new lumped structure photochemical mechanism for large-scale applications. Journal of Geophysical Research, 104(D23), 30387–30415. https://doi.org/10.1029/1999JD900876
- Zhang, Q., Zheng, Y., Tong, D., Shao, M., Wang, S., Zhang, Y., et al. (2019). Drivers of improved PM_{2.5} air quality in China from 2013 to 2017. Proceedings of the National Academy of Sciences of the United States of America, 116(49), 24463–24469. https://doi.org/10.1073/ pnas.1907956116
- Zhu, J., & Liao, H. (2016). Future ozone air quality and radiative forcing over China owing to future changes in emissions under the Representative Concentration Pathways (RCPs). Journal of Geophysical Research: Atmospheres, 121(4), 1978–2001. https://doi.org/10.1002/2015JD023926

References From the Supporting Information

- An, J., Huang, Y., Huang, C., Wang, X., Yan, R., Wang, Q., et al. (2021). Emission inventory of air pollutants and chemical speciation for specific anthropogenic sources based on local measurements in the Yangtze River Delta region, China. Atmospheric Chemistry and Physics, 21(3), 2003–2025. https://doi.org/10.5194/acp-21-2003-2021
- Brauner, N., & Shacham, M. (1997). Statistical analysis of linear and nonlinear correlation of the Arrhenius equation constants. *Chemical Engineering and Processing: Process Intensification*, 36(3), 243–249. https://doi.org/10.1016/S0255-2701(96)04186-4
- Calvin, K., Bond-Lamberty, B., Clarke, L., Edmonds, J., Eom, J., Hartin, C., et al. (2017). The SSP4: A world of deepening inequality. *Global Environmental Change*, 42, 284–296. https://doi.org/10.1016/j.gloenycha.2016.06.010
- DeMore, W. B., Sander, S. P., Golden, D. M., Hampson, R. F., Kurylo, M. J., Howard, C. J., et al. (1997). Chemical kinetics and photochemical data for use in stratospheric modeling, evaluation number 12. JPL publication 97-4.
- Duan, H., Zhou, S., Jiang, K., Bertram, C., Harmsen, M., Kriegler, E., et al. (2021). Assessing China's efforts to pursue the 1.5°C warming limit. Science, 372(6540), 378–385. https://doi.org/10.1126/science.aba8767
- Emmons, L. K., Schwantes, R. H., Orlando, J. J., Tyndall, G., Kinnison, D., Lamarque, J. F., et al. (2020). The chemistry mechanism in the Community Earth System Model version 2 (CESM2). Journal of Advances in Modeling Earth Systems, 12(4), 1–21. https://doi. org/10.1029/2019MS001882
- Hong, S. Y., Noh, Y., & Dudhia, J. (2006). A new vertical diffusion package with an explicit treatment of entrainment processes. *Monthly Weather Review*, 134(9), 2318–2341. https://doi.org/10.1175/MWR3199.1
- Huang, L., Zhu, Y., Zhai, H., Xue, S., Zhu, T., Shao, Y., et al. (2021). Recommendations on benchmarks for numerical air quality model applications in China—Part 1: PM_{2.5} and chemical species. *Atmospheric Chemistry and Physics*, 21(4), 2725–2743. https://doi.org/10.5194/ acp-21-2725-2021
- Iacono, M. J., Delamere, J. S., Mlawer, E. J., Shephard, M. W., Clough, S. A., & Collins, W. D. (2008). Radiative forcing by long-lived greenhouse gases: Calculations with the AER radiative transfer models. *Journal of Geophysical Research*, 113(13), 2–9. https://doi. org/10.1029/2008JD009944
- Kuik, F., Lauer, A., Churkina, G., Denier Van Der Gon, H. A. C., Fenner, D., Mar, K. A., & Butler, T. M. (2016). Air quality modelling in the Berlin-Brandenburg region using WRF-Chem v3.7.1: Sensitivity to resolution of model grid and input data. *Geoscientific Model Development*, 9(12), 4339–4363. https://doi.org/10.5194/gmd-9-4339-2016
- Lamarque, J. F., Emmons, L. K., Hess, P. G., Kinnison, D. E., Tilmes, S., Vitt, F., et al. (2012). CAM-chem: Description and evaluation of interactive atmospheric chemistry in the Community Earth System Model. *Geoscientific Model Development*, 5(2), 369–411. https://doi. org/10.5194/gmd-5-369-2012
- Morrison, H., Thompson, G., & Tatarskii, V. (2009). Impact of cloud microphysics on the development of trailing stratiform precipitation in a simulated squall line: Comparison of one- and two-moment schemes. *Monthly Weather Review*, 137(3), 991–1007. https://doi.org/10.1175/2008MWR2556.1
- NCEP (2015). NCEP GDAS/FNL 0.25 degree global tropospheric analyses and forecast grids. Research Data Archive at the National Center for Atmospheric Research, Computational and Information Systems Laboratory. https://doi.org/10.5065/D65Q4T4Z
- Sander, S. P., Friedl, R. R., Ravishankara, A. R., Golden, D. M., Kolb, C. E., Kurylo, M. J., et al. (2003). Chemical kinetics and photochemical data for use in atmospheric studies, evaluation number 14. JPL publication 02-25.
- Sicard, P., Crippa, P., De Marco, A., Castruccio, S., Giani, P., Cuesta, J., et al. (2021). High spatial resolution WRF-Chem model over Asia: Physics and chemistry evaluation. Atmospheric Environment, 244, 118004. https://doi.org/10.1016/j.atmosenv.2020.118004

- Sun, L., Cao, L., Ding, H., Gao, M., Li, S., & Chen, G. (2022). Influence of using different chemical mechanisms on simulations of ozone and its precursors in the troposphere of Shanghai, China. *Atmospheric Environment*, 289, 119299. https://doi.org/10.1016/j.atmosenv.2022.119299 Tan, J., Zhang, Y., Ma, W., Yu, Q., Wang, J., & Chen, L. (2015). Impact of spatial resolution on air quality simulation: A case study in a highly industrialized area in Shanghai, China. *Atmospheric Pollution Research*, 6(2), 322–333. https://doi.org/10.5094/APR.2015.036
- Tao, Z., Chin, M., Gao, M., Kucsera, T., Kim, D., Bian, H., et al. (2020). Evaluation of NU-WRF model performance on air quality simulation under various model resolutions—An investigation within the framework of MICS-Asia Phase III. *Atmospheric Chemistry and Physics*, 20(4), 2319–2339. https://doi.org/10.5194/acp-20-2319-2020
- Tian, R., Ma, X., Jia, H., Yu, F., Sha, T., & Zan, Y. (2019). Aerosol radiative effects on tropospheric photochemistry with GEOS-Chem simulations. Atmospheric Environment, 208, 82–94. https://doi.org/10.1016/j.atmosenv.2019.03.032
- Wild, O., Zhu, X., & Prather, M. J. (2000). Fast-J: Accurate simulation of in- and below-cloud photolysis in tropospheric chemical models. *Journal of Atmospheric Chemistry*, 37(3), 245–282. https://doi.org/10.1023/A:1006415919030
- Zheng, H., Cai, S., Wang, S., Zhao, B., Chang, X., & Hao, J. (2019). Development of a unit-based industrial emission inventory in the Beijing-Tianjin-Hebei region and resulting improvement in air quality modeling. *Atmospheric Chemistry and Physics*, 19(6), 3447–3462. https://doi.org/10.5194/acp-19-3447-2019
- Zhou, Y., Zhao, Y., Mao, P., Zhang, Q., Zhang, J., Qiu, L., & Yang, Y. (2017). Development of a high-resolution emission inventory and its evaluation and application through air quality modeling for Jiangsu Province, China. Atmospheric Chemistry and Physics, 17(1), 211–233. https://doi.org/10.5194/acp-17-211-2017
- Zhu, J., Chen, L., Liao, H., Yang, H., Yang, Y., & Yue, X. (2021). Enhanced PM_{2.5} decreases and O₃ increases in China during COVID-19 lockdown by aerosol-radiation feedback. *Geophysical Research Letters*, 48(2). https://doi.org/10.1029/2020GL090260